Research Paper



Is Online Exercise at Home More Effective Than Hydrotherapy and Physiotherapy in Patients With Non-specific Chronic Low Back Pain? A Randomized Clinical Trial

Mohammad Bayattork1* (0), Arash Khaledi2 (0)

Department of Sports Sciences and Physical Education, Faculty of Humanities Science, University of Hormozgan, Bandar Abbas, Iran.
Department of Sport Sciences, Kish International Campus, University of Tehran, Kish, Iran.



Citation Bayattork M, Khaledi A. Is Online Exercise at Home More Effective Than Hydrotherapy and Physiotherapy in Patients With Non-specific Chronic Low Back Pain? A Randomized Clinical Trial. Physical Treatments. 2022; 12(1):51-60. http://dx.doi.org/10.32598/ptj.12.1.517.1

doj http://dx.doi.org/10.32598/ptj.12.1.517.1



Article info: Received: 10 Aug 2021 Accepted: 23 Oct 2021 Available Online: 01 Jan 2022

ABSTRACT

Purpose: Many non-pharmaceutical methods have been proposed for the treatment of non-specific chronic low back pain (NCLBP), including online exercise at home (OEH), hydrotherapy, and physiotherapy approaches that have shown significant effects. Nevertheless, there are ambiguities in choosing the best option. Therefore, the present study was designed to compare these methods.

Methods: This randomized clinical trial included 60 patients with NCLBP (25-45 yrs). After selection, they were randomly divided into three groups (20 patients in each) of OEH (including core stability exercises and education), hydrotherapy, and physiotherapy (including hot pack, ultrasound, and TENS), and then, received interventions for 12 weeks. The Visual Analog Scale (VAS) was used to evaluate pain before and after treatment. The ANOCOVA and paired t-test were used to analyze the data and a significant level of P<0.05.

Results: The results showed that all three treatments used in this study had a significant effect (P<0.05) on reducing the pain intensity of the patients after the intervention. There was no significant difference between the three treatments in reducing pain; however, the OEH method could reduce the pain to a greater extent (32.79%).

Conclusion: Although the present study reaffirms the effectiveness of all three treatment approaches, no significant differences were found in the selection of the best option. However, depending on the patient's condition, one of these methods can be selected. Therefore, we introduce hydrotherapy, and physiotherapy methods for the elderly, severe pain and disability, and OEH as an available method to save money, and time and, most importantly, prevent the COVID-19.

Keywords:

Chronic low back pain, Online, Home exercise, Hydrotherapy, Physiotherapy

* Corresponding Author: Mohammad Bayattork Address: Sport Sciences and Physical Education Department, Faculty of Humanities Sciences, University of Hormozgan, Bandar Abbas, Iran. Phone: +98 (915) 5521005 E-mail: mohammadbayat82@gmail.com

Highlights

• All three treatment approaches are significantly effective in reducing pain.

• OEH, hydrotherapy, and physiotherapy were effective in reducing the VAS score, respectively; however, this difference was not significant.

• Saving economic costs and time as well as ease of execution are the advantages of home exercise over physiotherapy and hydrotherapy.

• OEH is an available and safe option to prevent COVID-19 in patients whit NCLBP.

Plain Language Summary

Physiotherapy and exercise therapy in water and land have always been the most popular and common non-pharmacological strategies for the treatment of non-specific low back pain. The high cost of treatment as well as spending a lot of time going to medical centers for a long time and most importantly the emergence of COVID-19 in the world have created many problems for these patients. These factors have even caused frustration and discouragement for many of these patients from pursuing treatment processes as well as improving low back pain. Therefore, we decided to compare home exercise with online monitoring against the two popular approaches of physiotherapy and hydrotherapy. Our results showed that all three methods are effective and almost equal in reducing pain intensity. However, online exercise at home was more prominent and covered all the problems mentioned above.

1. Introduction

ow back pain (LBP) after a cold is the second most common reason for seeing a doctor in the world [1] and is one of the most well-known musculoskeletal problems that usually affect everyone during their lifetime and has become a major socio-

economic problem [2]. Also, literature shows that one of the most important causes of disability is LBP, and despite its high prevalence, in many cases, the source of pain has not been proven, and therefore, the term nonspecific LBP has been used for it [3]. Nonspecific chronic low back pain (NCLBP) is pain minimum for three months, located above the inferior gluteal folds and below the costal margin without leg pain [4]. NCLBP is a major public health problem in industrialized societies, with a 60 to 80% prevalence [5, 6]. Different methods can be applied for the treatment and management of NCLBP, including pharmaceutical and non-pharmaceutical [7].

However, long-term use of the drug has side effects and after the cessation of treatment due to the lack of muscle imbalances, there is also the possibility of recurrence of pain [8]. Many non-pharmacological methods have been proposed, among which exercise therapy [9], hydrotherapy [7], and physiotherapy [10] are the most popular strategies among recent researchers and therapists, and their positive effects in improving pain have been proven. In a systematic review and meta-analysis, the efficiency of various exercises, including Pilates, stability/motor control, resistance, and aerobics training, were investigated and reported they are effective in treating NCLBP [11]. Moreover, Bai et al., in a review and meta-analysis study, pointed to the positive benefits of water exercise [7]. In another review article, Haile et al. reported that physiotherapy methods also effectively reduce LBP [12].

Although exercise therapy is the most common type of treatment [13, 14], recent studies have pointed to stability exercises as one of the best training approaches [1, 15, 16]. In a recent systematic review, this approach was introduced as the best solution among other exercises in LBP treatment [1]. The San Francisco Spine Institute is one of the first institutions to recommend dynamic back stabilization as a more effective treatment [17, 18]. This exercise has shown tremendous effects, relying on the major strengthening of the multifidus, paraspinal, pelvic floor, diaphragm, abdominal, and gluteal muscles [1, 19].

In contrast, medical experience has shown that waterbased exercise has beneficial results for musculoskeletal disorders, especially for patients with severe pain [20]. Three important properties of water, including buoyancy, resistance, and water flow, have made hydrotherapy effective in reducing pain. Research shows that hydrotherapy can positively affect pain, strength, muscle flexibility, performance, self-confidence, fitness, and stress reduction in disorders, such as LBP [10]. Furthermore, physiotherapy methods, including using ultrasound devices [12], Transcutaneous Electrical Nerve Stimulation (TENS) [21], and hot water packs [22], have been old and very common applications in reducing patients' back pain. Ultrasound and TENS devices, through surface waves and electrical stimulation with the help of a hot water bag, increase blood circulation in muscles and soft tissues, which is quite effective in improving the healing process and reducing the transmission of pain nerve commands [12, 21].

Nowadays, due to the availability of online communication networks and on the other hand, there are some urban problems such as traffic, limited access to physiotherapy and exercise therapy clinics, lack of time, and most importantly, the emergence of COVID-19 and its potential risks, which highlights online exercises at home (OEH) [23]. Although studies have found the patient's presence with a specialist more effective [24], the above reasons led us first to identify one of the best exercise methods recommended by researchers (stability exercise) [1, 15], and then try to implement this style of practice at home, and secondly, compare it with the two popular methods of physiotherapy and hydrotherapy. It should be noted that to cover the weaknesses of the absence of the specialist as much as possible, we decided to control the patients' exercises at home through video and face-to-face networks.

Generally, in the field of home exercise, especially after the onset of COVID-19, little research has been done to help patients with NCLBP. However, there is insufficient evidence of the effectiveness of this type of exercise in particular and its superiority over other popular methods (physiotherapy and hydrotherapy) in general [5, 20, 25]. Recently, researchers in the reviews and meta-analyses also have emphasized the need to conduct research comparing different training methods in order to more accurately identify the effects of each [11, 24].

Therefore, the importance of exercise at home, given the current situation, specially COVID-19, and since to our knowledge to date, no study has compared these three treatments, made it necessary for us to conduct the present study. Hence, the present study was designed to compare the effects of three different treatment approaches (namely physiotherapy, hydrotherapy, and OEH) among patients with NCLBP. We hypothesized that the OEH is more effective than other methods.

2. Methods

The current study was a single-blind (therapist and specialist) randomized clinical trial of 60 patients with NCLBP with a mean age of 25-45 years. The participants were selected by convenience sampling method. In addition, an independent research coordinator from a local physiotherapy and corrective exercises clinic in Tehran, Iran recruited them. All participants filled out written informed consent forms before inclusion, and a local ethics committee approved the study. All subjects were examined by a physician and referred to the clinic to treat LBP. The study was registered on the UMIN_RCT website, and the unique trial number is UMIN000046358.

The inclusion criteria were a history of pain for at least three months, a minimum score of two on the visual analog scale (VAS), willingness to participate in any randomly selected treatment groups, and receiving full doses of the COVID-19 vaccine. The exclusion criteria were the presence of the lumbar disc herniation, acute stage of lumbar disc protrusion, systematic diseases (such as cardiovascular), or any condition, which contraindicated. Spinal surgery, fractures or inflammatory diseases of the spine, or any pathophysiological disorder, body mass index (BMI) more than normal level, having any intervention in the last two months, taking medical drugs, and failure to participate in the treatment sessions [22].

Outcome measure

VAS was used to measure pain intensity. It is a 10 cm ruler with zero (no pain) and 10 (most severe pain). Subjects are asked to select a point according to their pain level. Finally, patients' pain scores were recorded before and after treatment interventions. Validity of 0.70 and reliability of 0.97 have been reported for this scale [26].

Treatment groups

Participants were randomized followed by a concealed allocation ratio of 1:1 by unlocking an opaque and sealed envelope, and a card inside demonstrated the group, to which the subjects were randomly allocated: OEH (n=20), hydrotherapy (n=20), and physiotherapy (n=20). Before starting the project, all participants were asked to complete the written consent form, and they were informed about the 12 weeks of clinical interventions and study procedure.

Online Exercise at Home

The first group performed OEH. They were given five exercises (Table 1). in addition, they received a manual guideline containing exercises and training on LBP, how to create it, and correct postural daily positions, including the proper technique of movement patterns, such as lifting objects off the ground, sitting, sleeping, walking as well as ergonomic methods at work and home [27]. They were asked to read this manual and follow it daily. The exercise was performed three times a week for 12 weeks. Each 20-30-minute exercise session included 5 minutes of cooldown and warm-up and the main program [28]. Further, the exercises were monitored online so that whenever the subjects were ready, they communicated with the experts using a computer or mobile phone through the videoconferencing software (https://meet.jit.si/) [29].

Hydrotherapy exercises

The second group performed their hydrotherapy (five exercises) in a pool (with a normal water temperature range of 30-32° C and 120-cm deep) under the supervision of a specialist (Table 2). This group had 40-50 minute exercise sessions three times a week for 12 weeks, including 15 minutes of warm-up (walking exercises to adapt to the pool conditions), cool down (bilateral and general stretching), and about 25 minutes of the main

program. Water exercises were performed in groups of 6 to 7 participants per session due to the closer supervision of hydrotherapy specialists and the pool's small size. Moreover, the number of repetitions and sets were designed based on the amount of pain and the ability of patients from the first week to the last week based on the principles of overload [20].

Physiotherapy methods

The third group was treated in a physiotherapy clinic under the supervision of a physiotherapist. All supervised intervention programs were applied for 50 to 60 minutes a day, three times per week, for 12 weeks. The heating surface is applied with hot packages for 20 minutes in the low back region. The patients also used ultrasound devices (Chattanooga, Tennessee, USA) for 10 minutes, which operated at a frequency of 1 MHz and an intensity of 1.5 W/cm, and a converter head with an area of 5, BNR 1:5, and an ERA of 4 cm. Slow circular movements were applied to the paravertebral lumbar region

Table 1. Method of Online Core Stability Exercises at Home

Compositions		Methods		
Curl up		Train the abdominal (rectus and obliques) muscles and control the movement of the pel- vis. The person is placed with 90° knee flexion, then the hands are placed crosswise on the chest, and the upper torso is raised to 30° and returns to the first position. 4 sets × 10 rep.		
Side Bridge		Train the oblique muscles and the quadratus lumborum, as critical muscles in the spine's stability. The person tries to keep his body along a straight line so that only the forearm and the outer ankle of the foot are in contact with the ground. 4 set×15 sec.		
Bird dog		Strengthen the posterior and anterior lumbar muscles, especially the transverse abdomi- nis. The person is placed in a dog position and at the same time keeps the opposite arm and leg in full extension along the ground. 4 set×10 rep		
Bridging		Strengthen the muscles of the rectus spine and gluteal. The person is placed on the mat- tress with full knee flexion, then lifts the body off the ground, maintains this position for 3 sec, and returns to the first position. 4 set×15 rep.		
Pelvic tilt		Strengthen the anterior abdominal and gluteal muscles. The person sleeps in the supine position and puts the knee in flexion, then with the contraction of the relevant muscles, he/she places the back completely flat on the ground so that the arch of the lumbar region disappears and maintains this position. 4 set×15 sec.		

PHYSICAL TREATMENTS

by the head of the transducer. Finally, TENS; 30–40 Hz was applied for 20 min [22].

Statistical analysis

Statistical analysis was performed using IBM SPSS version 20 for Windows (SPSS Inc., Chicago, IL, USA); all variables were reported using the descriptive statistic (Mean, Standard Deviation). The analysis of variance (ANOVA) was used to compare individual characteristics between groups. One-way analysis of the covariance (AN-COVA) and the post hoc test were used to compare groups, both at baseline (pre-test) and follow-up (post-test). In these models, the outcome variable (pain intensity in post-test) was entered as dependent variables, the group was used as a fixed factor, while the pre-test was used as covariates. Assumptions of ANOCOVA were considered: normal distribution of residues and homogeneity of variances in groups.

Table 2. Method of Hydrotherapy Exercise

In addition, paired t-test was used to compare pain intensity in pre and post-test in every group separately. The significance level was set at P<0.05.

3. Results

Individual characteristics were compared between groups using the ANOVA test for scale data (e.g., age and height) and no significant difference was observed between the groups (P> 0.05), which is shown in Table 3.

In Table 4, the ANCOVA showed no significant differences between groups in VAS (F (2, 56)=1.39, P=0.258, η 2=0.14). Comparisons of mean pain intensity are shown in Figure 1. A significant reduction in pain intensity was observed after therapy intervention as follows: OEH (32.79%, P<0.001), HT (29.58%, P=0.006), and HT (22.01%, P=0.012).

Compositions		Methods			
Waking (forward, backward and side)		Counteract the dynamic resistance, strength, and stretching of the iliopsoas muscle and contralateral hip flexor muscles. The person walks to forwards, back- ward, right, and left. (Each direction for about 2 to 3 minutes).			
Bring knee to chest		Strengthen and stretch the muscles of the hip flexors and extensors and strength- en the rectus abdominis. In the supine position, the person grabs the handrail, floats in the water, extends the hip and knee joint, and returns to the first position (Return the knee to the chest). 3 set×20-30 rep.			
Twisting side to side		Active rotation exercise is to mobilize the hip joint and thoracic/lumbar spine and strengthen and stretch the external and internal oblique abdominis muscles. Place hands on the handrail, then place the knee and thigh joint at an angle of about 90° and rotate it to the right and left sides. 3 set×20-30 rep.			
Top elbow to knee		Strengthen the abdominal muscles, especially the inside and outside, and hip flexors and mobilization of the hip joint and spine. Place the legs on the floor of the pool, then, with flexion of the hip and upper extremities, bring one hand to the knee of the opposite side, and then do the opposite side. 3 set×20-30 rep.			
Leg kicks		To counteract the dynamic resistance and strengthen the hamstring and quadri- ceps muscles. The person leans against rails or walls and kicks in a way that exerts a force similar to the movement of a bicycle. 4 set×20-30 rep.			
Posterior pelvic tilt		Strengthens the abdominal and gluteal muscles and stretches the iliopsoas and quadratus muscles of the lumbar spine. The person leans back against the pool wall and tries to smooth the arch of the lumbar spine with the wall with an isometric contraction. 4 set×15 sec.			

PHYSICAL TREATMENTS

	Mean±SD				
Variables	Online Exercise at Home (n=20) Hydrotherapy (n=20)		Physiotherapy (n=20)	Р	
Age (y)	38.5±12.68	35.1±15.12	37.3±13.11	0.125	
Height, cm	171.5±20.77	167.8±24.14	173.1±20.20	0.078	
Weight, kg	68.7±8.14	70.2±7.10	69.8±9.44	0.455	
Body mass index, kg/m ²	23.4±1.81	24.9±1.50	23.3±1.41	0.314	

Table 3. Individual characteristics in three treatment groups

PHYSICAL TREATMENTS

PHYSICAL TREATMENTS

Table 4. Results of analysis of covariance for pain intensity

Source	Sum of Squares	df	Mean Square	F	Р	٩°²
pre	8.625	1	8.625	2.452	0.001	0.42
groups	9.755	2	4.877	1.386	0.258	0.14
Error	196.999	56	3.518			

4. Discussion

To our knowledge, this is the first study to compare OEH, hydrotherapy exercise, and physiotherapy in patients aged 25-45 years with NCLBP. The results showed that the pain intensity decreased significantly after the implementation of all three different treatment approaches. Although there was no significant difference between the groups, the outcome of the online core stability exercise at home was more prominent.

In the first hypothesis, our results are consistent with some studies in terms of reducing patients' pain through exercise at home [22, 24, 30]. On the other hand, some studies are inconsistent in terms of relative superiority over other approaches. Bronfort et al. compared the three methods of supervised exercise, spinal manipulation, and home exercise for 12 weeks and reported that exercise

under the supervision of specialists, manipulation, and home exercise, respectively, were more effective in reducing pain in these patients [27]. In another study, Schulz et al. compared home exercise program supervised exercise plus home exercise program, or spinal manipulative therapy at 12 weeks, and stated that pain intensity was decreased (30% to 40%) after treatment in all three methods with the largest difference (8%) favoring manipulative therapy and home exercise over home exercise alone [31]. Torstensen et al. evaluated the effect of three progressive therapies, physiotherapy, and personal exercises on the pain of these patients [10]. This agreement might be related to the similarity in the use of Williams exercises combined with lumbar stabilizers training that could effectively match the results of the two studies. In general, there are fundamental differences in the kinetics

and kinematics of our movement pattern exercises with

them. In most previous studies, home exercises have been



Figure 1. Comparison of mean pain intensity in pre-test and post-test OEH: Online Exercise at Home; HT: Hydrotherapy; PT: Physiotherapy.

PHYSICAL TREATMENTS

performed without online supervision, and this principle has been observed in our study as the first approach. On the other hand, due to the acceptable results of stability exercise [1, 16], we focused only on these exercises aimed at strengthening the lumbopelvic muscles isometrically and isotonically as a second approach, which is less seen in other studies. Furthermore, the term positions and performing the correct techniques of daily activities is a very important third approach that we have considered. Sahrmann et al. also confirmed this issue and stated that the main musculoskeletal disorders, such as CLBP are due to the implementation of incorrect daily movement techniques and their repetition over time [32].

However, hydrotherapy is the second effective and significant hypothesis in improving patients' symptoms after the home exercise approach from the numerical point of view, which does not show significant results between the three methods. From the inconsistent results, we can mention the study by Dundar et al. They showed that aquatic exercise interventions have more significant effects than home-based exercise, which is in stark contrast to our results [20]. The most important possible reason for the superiority of water training can be attributed to the lack of direct supervision of specialists in the land training group. Because in the follow-ups, patients shared their problems and physical condition with the therapists only once a week by phone [20]. Similar findings were repeated by Bello et al. after six weeks of training. However, in their study, the land-based exercise group was performed in the presence of a specialist [33]. The type of exercise protocol can be used to justify this finding; these researchers designed the type, dose, and movement patterns in water and land exercises are quite similar. On the other hand, Sami et al. could not find a significant difference in comparison between hydrotherapy, relaxation, and McKenzie exercise methods, while in their study, all three methods had significant effects on improving pain symptoms [18]. Our results are consistent with the reports of these researchers in terms of hydrotherapy effects [34].

The third hypothesis of the present study shows that physiotherapy is not superior but an effective method after home and water exercises in reducing patients' pain. Several researchers have repeated this finding in the past and are consistent with their reports [35, 36]. Cecchi et al. compared the three methods of spinal manipulation, trunk training, and physiotherapy. Their results showed that spinal manipulation exercises were better than the other two groups [35]. Noori et al. found that both exercise therapy and physiotherapy positively reduced the subjects' pain; however, no significant differences were observed between the two methods [36].

Nonetheless, in our study, the physiotherapy approach was numerically placed after exercise at home and hydrotherapy. However, this situation was not statistically significant, inconsistent with the study of Kumar et al. They tested physiotherapy modalities (ultrasound, shortwave diathermy) and exercise (lumbar strengthening) against dynamic stability exercises on hockey players with moderate pain. After four weeks, although the physiotherapy group received the usual exercises, the stability exercises group became more efficacious [37]. From this point of view, one of the possible reasons can be attributed to doing stability group exercises in the clinic, and on the other hand, not using TENS in the physiotherapy group is another possible reason.

Hence, in general, all three approaches we consider are effective in improving these patients' pain, but with an overview of the benefits and harms of each of these methods, the best option can be selected, which we will mention the most important ones below. If we pay attention to the research process, we will find that the time spent performing exercises at home was about 20 to 30 minutes. This time in physiotherapy and hydrotherapy is about 45 to 60 minutes, in addition to the time that a person goes to the clinic or pool also adds to the disadvantages of these two methods, which can even increase the pain and prolong the treatment process for the patient. On the other hand, the cost of treatment for treating patients in physiotherapy and swimming pools is much higher than practicing at home, which adds to the disadvantages of both approaches. It is necessary to mention that there is no set time for performing exercises at home and the persons can do these exercises according to their free time, and more importantly, after the epidemic of the COVID-19, the home environment is the safest place against public treatment places and swimming pools [24]. In contrast, doing stability exercise may not be possible for patients with severe pain, whereas physiotherapy and hydrotherapy will be preferred, respectively [22]. However, in order to minimize the recurrence of pain and the possibility of skeletal injury, we designed exercises in a closed movement chain.

In conclusion, the present study reaffirms the hypothesis of the effectiveness of all three methods (physiotherapy, hydrotherapy, and OEH) in treating these patients [7, 11, 12], and the superiority of one of these approaches could not provide significant results. However, due to the advantages and disadvantages mentioned above and, on the other hand, improved pain scores numerically on the VAS scale, OEH is introduced as the best option for the treatment of NCLBP.

This study has two weaknesses. First, minimal access to research that has done home exercises under online supervision has made it difficult to discuss our findings with this type of research. Second, the lack of a control group to more accurately diagnose the effects of each intervention over time is one of the most important weaknesses of our study because it is possible that patients' pain would have worsened without the use of treatment and management methods or vice versa.

In contrast, our study has strengths in several respects: 1), covering the most important weakness of exercise at home, despite the many benefits reported in previous studies, monitoring patients online [23, 24, 27, 30], 2) the use of stability exercises is one of the best therapeutic approaches globally for the home exercise group [1,11, 38]; 3) comparing the most popular and effective noninvasive strategies [11, 22, 33, 34] in a research design to help more accurately identify the effects of each for researchers, especially future review and meta-analysis; 4) blinding therapists or specialists in order to achieve more reliable results, and 5) using online exercises for the first time compared to physiotherapy and hydrotherapy methods. We suggest that future researchers use exercise-supervised online therapy to evaluate other side effects of LBP pain alongside the double-blind control group (patients and therapists).

5. Conclusion

The present study results indicated the effectiveness of all three approaches in reducing chronic pain in these patients. Although numerically, home exercises, hydrotherapy, and physiotherapy were ranked first to third, respectively, no significant difference was observed in the superiority of one of these methods. However, some important points could convince us to choose exercise at home.

Ethical Considerations

Compliance with ethical guidelines

All participants in this study have read and signed the consent form. Participants were informed of the process and purpose of the study and had the right to withdraw from the study at any time if they were not satisfied. In addition, if they wanted to know the results of this research, the information obtained was easily provided to them. The local ethics committee approved the whole research process.

Funding

The study did not receive any financial rewards from government, personal or commercial support organizations.

Authors' contributions

Data analysis and manuscript writing: Arash Khaledi; Data collection and manuscript correction: Mohammad Bayattork; Investigation, Mohammad Bayattork, and Arash Khaledi.

Conflict of interest

The authors declared no conflict of interest.

Acknowledgments

We thank all the specialists, therapists, participants, and officials from a local physiotherapy and corrective exercises clinic in Tehran who helped us in the process of conducting this research.

References

- [1] Khaledi A, Bayattork M, Gheitasi M. [The effectiveness of exercise therapy on improving pain and functional disability in patients with non-specific chronic low back pain: A systematic review of English clinical trials (Persian)]. Anesthesiology and Pain. 2020; 11(2):89-107. https://jap.iums.ac.ir/browse.php?a_id=5510&slc_lang=en&sid=1&printcase=1&hbnr=1&hbr=1
- [2] Oesch, P, Kool J, Hagen KB, Bachmann S. Effectiveness of exercise on work disability in patients with non-acute non-specific low back pain: Systematic review and meta-analysis of randomized controlled trials. Journal of Rehabilitation Medicine. 2010; 42(3):193-205. [DOI:10.2340/16501977-0524] [PMID]
- [3] Henchoz Y, Pinget C, Wasserfallen JB, Paillex, R, de Goumoëns P, Norberg M, et al. Cost-utility analysis of a threemonth exercise programme vs usual care following multidisciplinary rehabilitation for chronic low back pain. Journal of Rehabilitation Medicine. 2010; 42(9):846-52. [DOI:10.2340/16501977-0610] [PMID]
- [4] Chou R. Low back pain (chronic). BMJ Clinical Evidence. 2010; 2010:1116. [PMID] [PMCID]
- [5] Unsgaard-Tøndel M, Fladmark AM, Salvesen Ø, Vasseljen O. Motor control exercises, sling exercises, and general exercises for patients with chronic low back pain: A randomized controlled trial with 1-year follow-up. Physical Therapy. 2010; 90(10):1426-40. [DOI:10.2522/ptj.20090421] [PMID]
- [6] Macedo LG, Maher CG, Latimer J, McAuley JH. Motor control exercise for persistent, nonspecific low back pain: A systematic review. Physical Therapy. 2009; 89(1):9-25. [DOI:10.2522/ptj.20080103] [PMID]
- [7] Bai R, Macedo LG, Maher CG, Latimer J, McAuley JH. Effectiveness of spa therapy for patients with chronic low

back pain: An updated systematic review and metaanalysis. Medicine. 2019; 98(37):e17092. [DOI:10.1097/ MD.000000000017092] [PMID] [PMCID]

- [8] da Silva T, Mills K, Brown BT, Pocovi N, de Campos T, Maher C, Hancock MJ. Recurrence of low back pain is common: A prospective inception cohort study. Journal of Physiotherapy. 2019; 65(3):159-65. [DOI:10.1016/j.jphys.2019.04.010] [PMID]
- [9] Barros dos Santos AO, Pinto de Castro JB, Lima VP, da Silva EB, de Souza Vale RG. Effects of physical exercise on low back pain and cortisol levels: A systematic review with metaanalysis of randomized controlled trials. Pain Management. 2021; 11(1):49-57. [DOI:10.2217/pmt-2020-0020] [PMID]
- [10] Torstensen TA, Ljunggren AE, Meen HD, Odland E, Mowinckel P, Geijerstam S. Efficiency and costs of medical exercise therapy, conventional physiotherapy, and self-exercise in patients with chronic low back pain: A pragmatic, randomized, single-blinded, controlled trial with 1-year followup. Spine. 1998; 23(23):2616-24. [DOI:10.1097/00007632-199812010-00017] [PMID]
- [11] Owen PJ, Miller CT, Mundell NL, Verswijveren SJ, Tagliaferri SD, Brisby H, et al. Which specific modes of exercise training are most effective for treating low back pain? Network meta-analysis. British Journal of Sports Medicine. 2020. 54(21):1279-87. [DOI:10.1136/bjsports-2019-100886] [PMID] [PMCID]
- [12] Haile G, Hailemariam TT, Haile TG. Effectiveness of ultrasound therapy on the management of chronic non-specific low back pain: A systematic review. Journal of Pain Research. 2021; 14:1251. [DOI:10.2147/JPR.S277574] [PMID] [PMCID]
- [13] Ames J. Randomized controlled study of the outcome of hydrotherapy for subjects with low back or back and leg pain. The Journal of Aquatic Physical Therapy. 1999; 7(1):25. https://journals.lww.com/japt/Citation/1999/07010/A_ Randomized_Controlled_Study_of_the_Outcome_of.6.aspx
- [14] Soukup MG, Glomsröd B, Lönn JH, Bö K, Larsen S. The effect of a Mensendieck exercise program as secondary prophylaxis for recurrent low back pain: A randomized, controlled trial with 12-month follow-up. Spine. 1999; 24(15):1585-91. [DOI:10.1097/00007632-199908010-00013] [PMID]
- [15] Gomes-Neto M, Lopes JM, Conceição CS, Araujo A, Brasileiro A, Sousa C, et al. Stabilization exercise compared to general exercises or manual therapy for the management of low back pain: A systematic review and meta-analysis. Physical therapy in Sport. 2017; 23:136-42. [DOI:10.1016/j. ptsp.2016.08.004] [PMID]
- [16] Li X, Shen J, Liang J, Zhou X, Yang Y, Wang D, et al. Effect of core-based exercise in people with scoliosis: A systematic review and meta-analysis. Clinical Rehabilitation, 2021; 35(5):669-80. [DOI:10.1177/0269215520975105] [PMID] [PMCID]
- [17] Ferreira ML, Ferreira PH, Latimer J, Herbert R, Maher CG. Does spinal manipulative therapy help people with chronic low back pain? Australian Journal of Physiotherapy. 2002; 48(4):277-84. [DOI:10.1016/S0004-9514(14)60167-7] [PMID]
- [18] McKenzie RA, McKenzie R. The lumbar spine: Mechanical diagnosis & therapy. Rho: Spinal Publications. https://www.google.com/books/edition/The_Lumbar_ Spine/8AiAQgAACAAJ?hl=en

- [19] Lim EC, Poh RL, Low AY, Wong WP. Effects of Pilates-based exercises on pain and disability in individuals with persistent nonspecific low back pain: A systematic review with metaanalysis. Journal of Orthopaedic & Sports Physical Therapy. 2011; 41(2):70-80. [DOI:10.2519/jospt.2011.3393] [PMID]
- [20] Dundar U, Solak O, Yigit I, Evcik D, Kavuncu V. Clinical effectiveness of aquatic exercise to treat chronic low back pain: A randomized controlled trial. Spine. 2009; 34(14):1436-40. [DOI:10.1097/BRS.0b013e3181a79618] [PMID]
- [21] Yakşi E, Ketenci A, Baslo MB, Orhan EK. Does transcutaneous electrical nerve stimulation affect pain, neuropathic pain, and sympathetic skin responses in the treatment of chronic low back pain? A randomized, placebo-controlled study. The Korean Journal of Pain. 2021; 34(2):217-28. [DOI:10.3344/kjp.2021.34.2.217] [PMID] [PMCID]
- [22] Koldaş Doğan Ş, Sonel Tur B, Kurtaiş Y, Atay MB. Comparison of three different approaches in the treatment of chronic low back pain. Clinical Rheumatology. 2008; 27(7):873-81. [DOI:10.1007/s10067-007-0815-7] [PMID]
- [23] Shariat A, Anastasio AT, Soheili S, Rostad M. Home-based fundamental approach to alleviate low back pain using myofascial release, stretching, and spinal strengthening during the COVID-19 pandemic. Work. 2020; 67(1):11-9. [DOI:10.3233/WOR-203248] [PMID]
- [24] Quentin C, Bagheri R, Ugbolue UC, Coudeyre E, Pelissier C, Descatha A, et al. Effect of home exercise training in patients with nonspecific low-back pain: A systematic review and meta-analysis. International Journal of Environmental Research and Public Health. 2021; 18(16):8430. [DOI:10.3390/ ijerph18168430] [PMID] [PMCID]
- [25] Niemistö L, Rissanen P, Sarna S, Lahtinen-Suopanki T, Lindgren KA, Hurri H. Cost-effectiveness of combined manipulation, stabilizing exercises, and physician consultation compared to physician consultation alone for chronic low back pain: A prospective randomized trial with 2-year follow-up. Spine. 2005; 30(10):1109-15. [DOI:10.1097/01. brs.0000162569.00685.7b] [PMID]
- [26] Jensen MP, Chen C, Brugger AM. Interpretation of visual analog scale ratings and change scores: A reanalysis of two clinical trials of postoperative pain. The Journal of Pain. 2003; 4(7):407-14. [DOI:10.1016/S1526-5900(03)00716-8]
- [27] Bronfort G, Maiers MJ, Evans RL, Schulz CA, Bracha Y, Svendsen KH, et al. Supervised exercise, spinal manipulation, and home exercise for chronic low back pain: A randomized clinical trial. The Spine Journal. 2011; 11(7):585-98. [DOI:10.1016/j.spinee.2011.01.036] [PMID]
- [28] Gheitasi M, Khaledi A, Daneshjoo A. [The effect of combined core stability and sensory-motor exercises on Pain, Performance and movement fear in retired male athletes with non-specific chronic low back pain (Persian)]. Anesthesiology and Pain. 2020; 11(1):38-48. https://jap.iums.ac.ir/ browse.php?a_id=5483&sid=1&slc_lang=fa&ftxt=1
- [29] Yaghobi Tajani Z, Gheitasi M, Bayattork M, Andersen LL. Online supervised versus workplace corrective exercises for upper crossed syndrome: A protocol for a randomized controlled trial. 2021. [DOI:10.21203/rs.3.rs-408271/v1]
- [30] Kanas M, Faria RS, Salles LG, Sorpreso IC, Martins DE, Cunha RA, et al. Home-based exercise therapy for treating non-specific chronic low back pain. Revista da Associação

January 2022. Volume 12. Number 1

Médica Brasileira. 2018; 64:824-31. [DOI:10.1590/1806-9282.64.09.824] [PMID]

- [31] Schulz C, Evans R, Maiers M, Schulz K, Leininger B, Bronfort G. Spinal manipulative therapy and exercise for older adults with chronic low back pain: A randomized clinical trial. Chiropractic & Manual Therapies. 2019; 27(1):1-14. [DOI:10.1186/s12998-019-0243-1] [PMID] [PMCID]
- [32] Sahrmann S, Azevedo DC, Van Dillen L. Diagnosis and treatment of movement system impairment syndromes. Brazilian Journal of Physical Therapy. 2017; 21(6):391-9. [DOI:10.1016/j.bjpt.2017.08.001] [PMID] [PMCID]
- [33] Bello AI, Kalu NH, Adegoke BO, Agyepong-Badu S. Hydrotherapy versus land-based exercises in the management of chronic low back pain: A comparative study. Journal of Musculoskeletal Research. 2010; 13(04):159-65. [DOI:10.1142/S0218957710002594]
- [34] Sami S, Hakimi M, Ali-Mohammadi M, Karimiyani N. [Comparing the effects of hydrotherapy, relaxation and McKenzie exercise on improvement of chronic low back pain in athletes (Persian)]. Anesthesiology and Pain. 2014; 4(4):11-21. https:// jap.iums.ac.ir/files/site1/user_files_3d1973/daghigh-A-10-165-7-941efc4.pdf
- [35] Cecchi F, Molino-Lova R, Chiti M, Pasquini G, Paperini A, Conti AA, Macchi C. Spinal manipulation compared with back school and with individually delivered physiotherapy for the treatment of chronic low back pain: A randomized trial with one-year follow-up. Clinical Rehabilitation. 2010; 24(1):26-36. [DOI:10.1177/0269215509342328] [PMID]
- [36] Noori S, Ghasemi G, Khayambashi K, Karimi A, Minasian V, Alizamani S. Effect of exercise therapy and physiotherapy on patients with chronic low back pain. Journal of Isfahan Medical School. 2011; 29(151):1-7. https://web.s.ebscohost.com/abstract?direct=true&profile=ehscope=sth8005094
- [37] Kumar S, Sharma VP, Negi MP. Efficacy of Dynamic Muscular Stabilization Techniques (DMST) over conventional techniques in rehabilitation of chronic low back pain. The Journal of Strength & Conditioning Research. 2009; 23(9):2651-9. [DOI:10.1519/JSC.0b013e3181b3dde0] [PMID]
- [38] Wang XQ, Zheng JJ, Yu ZW, Bi X, Lou SJ, Liu J, et al. A meta-analysis of core stability exercise versus general exercise for chronic low back pain. PloS One. 2012; 7(12):e52082. [DOI:10.1371/journal.pone.0052082] [PMID] [PMCID]